Author's response to reviews

Title: Use of selective serotonin reuptake inhibitors and risk of re-operation due to post-surgical bleeding in breast cancer patients: a Danish population-based cohort study

Authors:

Rune Gärtner (rune.gartner@rh.regionh.dk)
Deirdre Cronin-Fenton (dc@dce.au.dk)
Heidi H. Hundborg (hh@dce.au.dk)
Lars Pedersen (lap@dce.au.dk)
Timothy L. Lash (tlash@bu.edu)
Henrik T. Sørensen (hts@dce.au.dk)
Niels Kroman (kroman@dadlnet.dk)

Version: 2 Date: 26 August 2009

Author's response to reviews:

Dear Editor,

Please consider the manuscript: “Use of selective serotonin reuptake inhibitors and risk of re-operation due to post-surgical bleeding in breast cancer patients: a Danish population-based cohort study” for publication in BMC Surgery.

This is the first population-based cohort study of selective serotonin reuptake inhibitors and risk of post-surgical bleeding in breast cancer patients. Data on risk of bleeding in context with surgical procedures and selective serotonin reuptake inhibitors are conflicting, but might have important clinical implications and calls for well sized studies to clarify the question.

All authors have contributed significantly to the work and have seen and approved the manuscript. The present Manuscript has not been published previously in print or electronic format and is not under consideration by another publication or electronic medium. No material submitted as part of the manuscript infringes existing copyrights, or the rights of a third party. The authors are understood to give BMC Surgery permission to reproduce the article or portions of it for print and online distribution.

None of the authors have published or submitted any related papers from the present study.

None of the authors have any potential financial or non-financial competing interests related to this manuscript.

We hope that this manuscript is acceptable for publication in BMC Surgery. Please do not hesitate to contact me if you have any questions or require more information.

Yours sincerely,
Henrik Toft Sørensen
Corresponding/submitting author